**Author's response to reviews**

**Title:** Utility and feasibility of integrating pulse oximetry into the routine assessment of young infants at primary care clinics in Karachi, Pakistan: a cross-sectional study

**Authors:**

Connor A Emdin (cemdin@georgeinstitute.org.uk)
Fatima Mir (fatima.mir@aku.edu)
Shazia Sultana (shazia.sultana@aku.edu)
A M Kazi (momin.kazi@aku.edu)
Anita KM Zaidi (anita.zaidi@aku.edu)
Michelle C Dimitris (michelle.dimitris@sickkids.ca)
Daniel E Roth (daniel.roth@sickkids.ca)

**Version:** 5  
**Date:** 28 August 2015

**Author's response to reviews:** see over
Dear Dr Shally Awasthi

Re: Utility and feasibility of integrating pulse oximetry into the routine assessment of young infants at primary care clinics in Karachi, Pakistan: a cross-sectional study

In response to the editors’ requests for clarification of the manuscript, we have clarified that written informed consent was obtained from all parents/guardians of infants and have added in the following sentence to the manuscript (page 5): “Written informed consent was obtained from all parents/guardians.”

Thank you for your consideration of the manuscript, and we look forward to hearing from you.

Yours sincerely,

Connor Emdin     Daniel Roth